

# ORCA - Online Research @ Cardiff

This is an Open Access document downloaded from ORCA, Cardiff University's institutional repository:https://orca.cardiff.ac.uk/id/eprint/138580/

This is the author's version of a work that was submitted to / accepted for publication.

Citation for final published version:

Zepeng, Wu, Mohammed, Ahmed and Harris, Irina 2021. Food waste management in the catering industry: enablers and interrelationship. Industrial Marketing Management 94, pp. 1-18. 10.1016/j.indmarman.2021.01.019

Publishers page: http://dx.doi.org/10.1016/j.indmarman.2021.01.019

Please note:

Changes made as a result of publishing processes such as copy-editing, formatting and page numbers may not be reflected in this version. For the definitive version of this publication, please refer to the published source. You are advised to consult the publisher's version if you wish to cite this paper.

This version is being made available in accordance with publisher policies. See http://orca.cf.ac.uk/policies.html for usage policies. Copyright and moral rights for publications made available in ORCA are retained by the copyright holders.



# Food waste management in the catering industry: enablers and interrelationships

## Abstract

Food waste has a wide range of negative implications on the environment, economy and society and it is a concern for the catering industry and a growing percentage of the population. This work investigates the main root causes of food waste, in the catering industry, based on data collected from 32 businesses located in Wales, UK. The research uses multi-criteria decision making methods (Analytic Hierarchy Process and Decision Making Trial and Evaluation Laboratory), to analyse the degrees of importance and interrelationships of food waste enablers using holistic cause and effect diagrams and priority ranking. The analysis identifies critical food waste enablers in different dimensions (procurement and storage, processing, operation strategy, people and consumer) for various business outlets. Those causes can be addressed with limited resources and create the maximum value for the business and environment. The methodology could be used as a strategic tool by businesses in daily operations by managing and controlling identified root causes.

# 1. Introduction

The restaurants and related services in the UK accounted for over half of the hospitality industry's employment with over 1.4 million jobs in 2014 (Oxford Economics, 2015). These services generated to the UK economy over £24bn of Gross Value Added (GVA) and contributed over 40% of the hospitality industry GVA with a turnover of £57bn (ibid., page 17). The industry has experienced significant growth over the last ten years, and food waste is one of key priorities that requires urgent attention. FAO and International Food Waste Coalition (2018) reported that the total amount of wasted food is approximately 1.3 billion tons and it is expected to increase to 3.4 billion tons annually due to the increase in population as reported by the World Bank in 2018. According to WRAP (2017), 75% of all food waste could have been avoided, where on average 66% of waste is attributed to food spoilage and preparation, and 34% is linked to customer leftovers. The cost of UK food waste in the hospitality sector in 2011 was estimated at £2.5 billion and restaurants, pubs and hotels account for 54% of this financial cost (WRAP, 2017). This study focuses on food waste in the catering industry that includes restaurants, bars, takeaway service, cafe and other business models and it involves any waste generated in facilities in pre-consumer and postconsumer phases (Baldwin, 2012). Although some catering businesses introduced clear policies to prevent food waste (Bharucha, 2018), several researchers suggested that those challenges have not been resolved and need more focused attention (Pirani & Arafat, 2016; Sakaguchi et al., 2018). Furthermore, during the COVID-19 pandemic the industry had to deal with unprecedented challenges (Gursoy and Chi, 2020) and food waste appeared high in the agenda (Smithers, 2020). Due to last minute announcement of lockdowns in different parts of UK, businesses struggled to plan their stock levels to accommodate ever-changing rules that contributed further to food waste issues. This is in addition to a decline in dining, the introduction of strict hygiene and curfew times (e.g. 10 pm is some areas).

Food waste refers to the food that does not include animal feed, inedible parts of products; and originally produced for human consumption and is wasted somewhere in the food supply chain (FAO, 2013; Gustavsson et al., 2011). Figure 1 illustrates the different stages of a supply chain where food waste could happen. For perishable products, the cooling equipment is very important (Bilska et al., 2016) and strict regulations on food quality and safety force suppliers to dispose of food that is not suitable for consumption (Mena et al., 2014). At distributors, wholesalers and/or retailers, food waste results from operational enablers that include inappropriate packaging, mishandling, transportation and poor ordering policies (Mena et al., 2011; Lukic et al., 2014; Cicatiello et al., 2016) and the expiration date is another common reason (Lewis et al., 2017). At the household stage, consumerhuman enablers are the most critical due to a lack of understanding of food waste issues (Buzby et al., 2011; Gustavsson et al., 2011), over-stocking or confusion over "use by" and "best before" dates (Parfitt et al., 2010; Gustavsson et al., 2011). In catering, food waste occurs when food is prepared in the kitchen and consumed by customers (Baldwin, 2012). In the commercial kitchen, the core contributor could be preparation residue from unskilled trimming process or over production of food that exceeds the customer demand (Baldwin, 2012; Papargyropoulou et al., 2016). The food waste from customers mainly occurs as a result of leftovers due to several reasons, such as bad flavour, avoiding certain food categories and/or large portion sizes (BCFN, 2012; Santos, 2017).



Figure 1. Food waste and different stages of the food supply chain (adapted from Gustavsson et al., 2011; Mena et al., 2011; BCFN, 2012; Priefer et al., 2016).

Sustainable food waste management leads to a reduction of greenhouse gas emissions, which in turn gives possible economic gains. Sustainability is a topic that is high on the agenda in all disciplines including industrial marketing (Oruezabala & Rico, 2012). Wind (2006) reflected on the evolution of the industrial marketing field over 40 years and emphasized the rise of small businesses, importance of breaking functional silos and that developments on the Internet led to the convergence of B2B and B2C markets. The research presented in this paper links all parts of the value chain where there is an on-going interaction between functions and different stakeholders. Through the integration of marketing with sustainable supply chain functions, and the use of effective communication, customers and other stakeholders (e.g. businesses) will become informed of the sustainable practices initiated by companies (Chan et al., 2012; Oruezabala & Rico, 2012). The strategic tool presented in this paper would help businesses to assess the food waste enablers that should be targeted as a priority to support the on-going development of their positive corporate image and sustainable agenda including their marketing strategy presented to other businesses. This research would lead to the development of food waste reduction approaches with advantages to marketing strategies in B2B and B2C environments.

The causes of food waste have many reasons, thus identifying and reducing the main causes requires substantial attention (Papargyropoulou et al. 2016). Researchers and practitioners are increasingly interested in mitigating avoidable food waste where possible. However, progress in this area has been slow to target the root causes of food waste and uncertain identification of root causes could lead to confusion regarding the order in which they should be targeted resulting in low efficiency. To design effective food waste prevention and reduction strategies it is critical to understand food waste causes and their interrelationships where the majority of published studies focus on a single type of the business model (e.g. Falasconi et al. 2015; Betz et al, 2015; Duursma et al., 2016, Bharucha et al., 2018; Filimonau et al. 2019a; Coşkun & Özbük, 2020). Filimonau & de Coteau (2019) and Filimonau et al. (2020a) emphasize that there is a lack of research that uses quantitative assessments of food waste enablers in the industry where published case studies, for example, of specific restaurants are not necessarily sector representative. They confirmed our findings that only very few papers used quantitative techniques to analyse the importance of food waste enablers. More recently, Parashar et al. (2020) used structural modelling and matrix of cross-impact multiplications to formulate food waste enablers across an entire supply chain towards the diminishment of carbon footprint level. Raut et al. (2018) performed an analysis study using AHP method on food waste causal factors at the post-harvest stage of fruits and vegetables supply chain.

Decision makers need to identify the main root causes of food waste to direct their efforts in mitigating and avoiding them rather than factors that include both causes and effects. Although some research conducted on causes of food waste (Mena et al., 2014; Parizeau et al., 2015; Bonadonna et al., 2019; and Aschemann-Witzel et al., 2020), their findings do not categorize them as root causes and effects to support better decision making (Filimonau, V. and Delysia, 2019; de Moraes et al., 2020), especially, in the catering sector. Furthermore, to the best of our knowledge, no research undertook cross-comparison analysis among business types and applied the multi-criteria decision making (MCDM) methods (i.e. Analytic Hierarchy Process (AHP) and Decision Making Trial and Evaluation Laboratory (DEMATEL) to explore the significance and interaction effect of these causes in the food waste context that this research considers. We also believe that this may be the first study to present a comprehensive literature review on food waste management in the catering industry and it develops holistic cause-effect diagrams using MCDM. Within these boundaries, this research seeks to contribute to the fields of food waste management and sustainable supply chain management via the multi-criteria decision making and it has links to other disciplines such as industrial marketing. The study would help decision makers to address the following research questions:

RQ1: What are the factors (enablers) of food waste in the catering industry?

RQ2: What are the root causes and effects of food waste out of those factors (enablers)?

RO3: Where should managers direct their efforts as a priority to mitigate and avoid them?

A hybrid AHP-DEMATEL method is proposed to identify the most important food waste enablers and the relationship among those enablers. The application of MCDM methods will help managers to (i) quantify the relative importance of food waste enablers (i.e. by using the AHP method); (ii) to clarify the interrelationships and correlation among these enablers and to categorize them as causes and effects (i.e., by using the DEMATEL method); (iii) to guide managers in prioritizing their efforts to address the causes rather than effects; and (iv) to target causes based on their revealed importance via AHP. This research develops a holistic cause-effect diagram and priority rank table of enablers in the general catering sector and specific food businesses (i.e. coffee shops, takeaways, restaurants and pubs). These points would not be achieved, in detail, without the application of MCDM methods considering (i) the amount of experts' opinion collected that might conflict where aggregation required, and (ii) a large number of food waste enablers to be evaluated. The advantage of MCDM is in an aggregation of decision-makers opinions when considering the multiple evaluation criteria (Shih, 2008). This cause-effect and relative importance analysis would allow decision makers to identify potential interventions among enablers and to help them to address these root food waste causes via tailored mitigation strategies that would boost sustainable food management in the future.

This paper continues as follows. In section 2, food waste literature is presented to summarise the major enablers of food waste in a unified framework. Section 3 presents the methodology, including AHP, DEMATEL and the questionnaire design. Section 4 presents the findings and Section 5 includes the discussion. Section 6 concludes with theoretical and managerial implications and future research directions.

#### 2. Literature Review

#### 2.1 Causes of food waste and waste minimisation strategies

This section discusses the causes of food waste (Table 1) and waste minimisation strategies (Table 2) that can be applied by businesses in the catering industry at different levels: procurement and storage, processing, operation strategy, people and consumer. We included the review of food waste reduction strategies because both topics (causes and strategies) are interlinked and this discussion informs the reader (especially business managers) of possible strategies and alternatives that could be available to them as a result of identifying the most significant food waste causes, and how possibly to target root causes that were identified as a priority in different levels. To identify these levels and factors, a review of the literature concerned with food waste was conducted using business information databases (Scopus, Google Scholar) and keywords such as "food waste", "food waste causes", "food waste strategies", "food waste multi-criteria decision making". Then a categorisation based on the literature review was undertaken where a spreadsheet was developed to classify all factors. A coding system was designed and agreed to facilitate identification of the common features using an iterative approach to highlight those factors that were coded into themes (e.g. levels) according to their features. As a result of the categorisation, the following levels were identified: procurement and storage, processing, operation strategy, people and consumer.

The procurement and storage level concerned with uncertainty related to customer demand and other enablers that are not under management control. Procurement and the storage of the raw materials is the most basic function of the business to maintain their supply to meet the demand. The waste occurs at this level as a result of that uncertainty and is well documented in the literature, which could lead to purchasing an excessive inventory of raw ingredients (Derqui et al., 2016). Demand management is critical here when balancing the needs of several stakeholder and customer groups (Brindley & Oxborrow, 2014). With regards to strategies, one option would be to make a prediction plan based on the historical data taking into account weather, season, holiday and other external enablers (Betz et al., 2015; Priefer et al., 2016). The forecasting of consumer demand as a prime strategy is discussed by Filimonau et al. (2020b) in their UK and Netherlands study, where restaurateurs in both countries

use this approach but forecasting does not always work and in-house training on forecasting and procurement for managers and chefs is highlighted (Filimonau & de Coteau, 2019).

The food processing level has a number of activities that transform the raw materials into food that is served to customers. At this level, the strict quality and safety regulations are the main drivers for the disposal of unqualified food (Charlebois & Hielm, 2014). Noticeably, preparation residue includes the waste that is generated from over-production, peeling, cutting, expiration, and trimming which could be regarded as a common reason for food waste in the commercial kitchen (Papargyropoulou et al., 2016). Businesses must reduce the preparation losses during processing, different strategies can avoid over trimming (Sakaguchi et al., 2018) through improvement in food preparation techniques (Papargyropoulou et al., 2016) and employee training (Betz et al., 2015). McAdams et al (2019) discuss that quality assurance standards are the most consistent key contributor towards food waste across the participating restaurants in their study.

Levels	Causes	References					
Δ	Difficulty in forecasting (A1)	BCFN (2012), Brindley & Oxborrow (2014), Charlebois et al. (2015), Derqui et al. (2016), Filimonau et al. (2020b), Gao et al. (2021), Priefer et al. (2016), Santos (2017)					
Procurement	Lack of purchase plan (A <sub>2</sub> )	Baldwin (2012), Santos (2017)					
and Storage	Forgotten and spoiled food (A <sub>3</sub> )	BCFN (2012), Charlebois et al. (2015), Derqui et al. (2016), Gao et al. (2021)					
	Over-stocking (A <sub>4</sub> )	Baldwin (2012), Charlebois et al. (2015), Derqui et al. (2016),					
	Strict regulation on safety and quality $(B_1)$	Baldwin (2012), Gao et al. (2021), Gustavsson et al. (2011), Papargyropoulou et al. (2016), McAdams et al (2019), Priefer et al. (2016), Santos (2017)					
B. Processing	Preparation residue (B <sub>2</sub> )	Baldwin (2012), BCFN (2012), Derqui et al. (2016), Filimonau & de Coteau (2019), Papargyropoulou et al. (2016), Santos (2017)					
	Unaware of cooking amount (B <sub>3</sub> )	BCFN (2012), Halloran et al. (2014)					
	Lack of equipment and tools (B <sub>4</sub> )	Rodgers (2005)					
C. Operation	Inappropriate portion size (C1)	BCFN (2012), Baldwin (2012), Betz et al. (2015), Charlebois et al. (2015), Derqui et al. (2016), Falasconi et al. (2015), Priefer et al. (2016), Lipinski et al. (2013), McAdams et al (2019), Santos (2017)					
strategy	Buffet style leading to over-taking $(C_2)$	Baldwin (2012), Gustavsson et al. (2011), Priefer et al. (2016)					
C. Operation strategy	Large menus (C <sub>3</sub> )	Charlebois et al. (2015)					
	Poor waste management (C <sub>4</sub> )	Derqui et al. (2016)					
	Poor layout of facility (C <sub>5</sub> )	Panisello & Quantick (2001)					
	Staff mishandling (D <sub>1</sub> )	Baldwin (2012), Charlebois et al. (2015)					
D. People	Miscommunication between management and staff $(D_2)$	Baldwin (2012), Charlebois et al. (2015), Papargyropoulou et al. (2016)					
_	Unskilled staff (D <sub>3</sub> )	Baldwin (2012), Charlebois et al. (2015); Filimonau et al. (2019a)					
	Attitude towards food waste (D <sub>4</sub> )	Gustavsson et al. (2011), WIE (2014)					
	Customer attitude, values and behaviours towards food $(E_1)$	Coşkun & Özbük (2020), Filimonau et al. (2019a), Gustavsson et al. (2011), Lipinski et al. (2013), Lorenz et al. (2017), Santos (2017), WIE (2014)					
E Consumer	Random customer purchase (E <sub>2</sub> )	Baldwin (2012), Charlebois et al. (2015); Filimonau et al. (2019a)					
E. Consumer	Different dietary preference of customers (E <sub>3</sub> )	Baldwin (2012), Betz et al. (2015)					
	Considering food unattractive $(E_4)$	Betz et al. (2015), Falasconi et al. (2015)					
	Over-ordering $(E_5)$	Santos (2017); Gao et al. (2021)					

Tał	ole	1. I	Food	waste	causes	in	the	food	servi	ice	sector.	•
-----	-----	------	------	-------	--------	----	-----	------	-------	-----	---------	---

The food waste produced at the operation strategy level is concerned with specific plans deployed by different food outlets related to their menu offerings, waste management and layout. Inappropriate portion size is discussed

by several academics as the most critical driver in the operation strategy level (Betz et al., 2015). Many studies proposed different methods for a better adaptation of portion size to customer needs, which includes offering different portions sizes (Sakaguchi et al., 2018), smaller portions (Coşkun & Özbük, 2020) and allowing customers to choose the size of serving (Bharucha, 2018), and buffet composition where a customer can decide on how much to take and what to eat (Priefer et al., 2016; Filimonau & de Coteau, 2019). Harvey et al (2019) also discuss that reduction in food waste can be achieved through the use of popular food sharing mobile applications that reconfigure traditional supply chain roles of consumers.

Levels	Possible strategies	References
	Measuring the occurring food waste	Clowes et al. (2018), Derqui et al. (2016), Duursma et al. (2016), Sakaguchi et al. (2018)
	Analysis of historical data, forecasting	Betz et al. (2015); Brindley & Oxborrow (2014), Derqui et al. (2016), Duursma et al. (2016), Filimonau & de Coteau (2019), Priefer et al. (2016)
A. Procurement and Storage	Optimisation of storage management	Betz et al. (2015), Pirani & Arafat (2016), Sakaguchi et al. (2018)
	Using the food near expiration date first	Betz et al. (2015), Bharucha (2018), Derqui et al. (2016), Filimonau et al. (2019a), Gao et al. (2021)
	Enhancement of order interval and reduction in stock pilling	Betz et al. (2015)
	Preparation losses control (e.g. avoid over trimming)	Betz et al. (2015), Sakaguchi et al. (2018), Papargyropoulou et al. (2016)
B. Processing	Development of strategies against overproduction	Betz et al. (2015), Clowes et al. (2018), Filimonau & de Coteau (2019)
D. Trocessing	Reusing the leftovers (under permission of law)	Betz et al. (2015), Clowes et al. (2018), Filimonau & de Coteau (2019)
	Fast cooling down of food to avoid microorganism growth	Betz et al. (2015)
A. Procurement and Storage B. Processing C. Operation strategy D. People E. Consumer	Adaption of portion size to customer needs	Betz et al. (2015), Clowes et al. (2018), Coşkun & Özbük (2020), Derqui et al. (2016), Duursma et al. (2016), Papargyropoulou et al. (2016), Priefer et al. (2016), Sakaguchi et al. (2018)
	Rethinking the buffet (e.g. using small serving bowls at buffet)	Betz et al. (2015), Bharucha (2018), Clowes et al. (2018), Derqui et al. (2016), Papargyropoulou et al. (2016)
	Possible donation of food to local charities	Betz et al. (2015), Bharucha (2018), Harvey et al (2019), Sakaguchi et al. (2018)
strategy	Redesign of supply chain to use existing food as possible	Sakaguchi et al. (2018)
A. Procurement and Storage         Measuring the occurring food waste         Clowes et al. (2015); Brindley 4 al. (2016), Duursma et al. (2 (2019), Priefer et al. (2015); Differ et al. (2015); Differ et al. (2016); Using the food near expiration date first         Betz et al. (2015); Brindley 4 al. (2018).           Using the food near expiration date first         Betz et al. (2015); Brindley 4 (2018).         Betz et al. (2015); Bharucha Filimonau et al. (2018).           B. Processing         Preparation losses control (e.g. avoid over trimming) Development of strategies against overproduction         Betz et al. (2015). Clowes et Coteau (2019).           B. Processing         Development of strategies against overproduction Reusing the leftovers (under permission of law)         Betz et al. (2015). Clowes et Coteau (2019)           Betz et al. (2015).         Betz et al. (2015).         Betz et al. (2015).           Rethinking the buffet (e.g. using small serving bowls at buffet)         Betz et al. (2015).         Betz et al. (2015).           Possible donation of food to local charities         Betz et al. (2015).         Barucha Sakaguchi et al. (2016).           Rethinking the buffet (e.g. using small serving bowls at buffet)         Betz et al. (2015).         Betz et al. (2015).           Possible donation of food to local charities         Sakaguchi et al. (2015).         Barucha Sakaguchi et al. (2016).           Rethinking the buffet by "pay by weight" system         Der(ef et al. (2015).         Barucha Sakaguchi et al. (2016).	Bharucha (2018)	
	Derqui et al. (2016)	
	Priefer et al. (2016)	
	Make food more attractive	Betz et al. (2015), Gao et al. (2021)
	Staff training	Betz et al. (2015), Bharucha (2018), Clowes et al. (2018), Derqui et al. (2016), Filimonau & de Coteau (2019), Gao et al. (2021), Priefer et al. (2016), Sakaguchi et al. (2018)
D. People	Better communication between staff and management to increase awareness about food waste	Betz et al. (2015), Clowes et al. (2018), Duursma et al. (2016), Pirani & Arafat (2016), Priefer et al. (2016), Sakaguchi et al. (2018)
	Enhancing coordination between the departments	Papargyropoulou et al. (2016)
E. Consumer	Communicating with customers to enhance awareness of food waste issues	Betz et al. (2015), Coşkun & Özbük (2020), Derqui et al. (2016), Duursma et al. (2016), Ellison et al. (2019), Filimonau et al. (2020a), Gao et al. (2021), Harvey et al (2019), Lipinski et al. (2013), Pirani & Arafat (2016), Sakaguchi et al. (2018)
	Collecting feedback to survey plate waste reasons	Betz et al. (2015)
	Rewarding sustainable behaviour	Duursma et al. (2016)

Table 2. Possible strategies for food waste reduction in the food service sector.

The people level concentrated on staff training and their attitude towards work and food waste with higher levels of management control. Poor communication between the management and staff can trigger staff mishandling, placing of wrong orders, spoiled and spilt food (Charlebois et al., 2015). To tackle this, for example, staff training as an effective way to control food waste has been discussed by several researchers. Through the training, staff can improve their skills (e.g. purchasing, storing and freezing) which is critical to food waste reduction (Priefer et al., 2016). Better communication is also a helpful way to engage staff in food waste prevention (Clowes et al., 2018) and increase staff awareness of waste (Sakaguchi et al., 2018).

At the consumer level, the food waste generated as a result of consumer behaviour and their attitudes (Filimonau et al. 2019b), needs, dietary requirements, and food appearance. The consumer here can be an individual (B2C) or business (B2B) and when dealing with, for example, public sector organisations (as a consumer), the business would also need to comply with their sustainable regulations (Oruezabala & Rico, 2012). The random customer purchase pattern and different customer dietary preferences and attitudes increase the challenge in forecasting the food quantity to produce (Baldwin, 2012). Some prevention strategies include raising awareness about food waste among customers where posters can be used to highlight ordering what you need (Priefer et al., 2016). Businesses can launch an awareness campaign to make customers realize the severity of food waste (Lipinski et al., 2013; Filimonau & de Coteau, 2019; Coşkun & Özbük, 2020).

#### 2.2 Multi-criteria decision making (MCDM) in food waste management

Multi-criteria decision making methods primarily used for choosing probable options using multiple criteria in a complex system (Mardani et al., 2015) and there are many techniques available, such as AHP, TOPSIS, ISM, MICMAC. AHP can be used to evaluate the most important food waste enablers; where some of them may have an impact on other major causes thereby influencing the strategies of food waste reduction. DEMATEL method can be used to identify the interrelationship among the enablers in the system (Tzeng et al., 2007) and divide those enablers into cause and effect groups (Wu & Tsai 2012). Table 3 presents the literature of MCDM methods that have been applied in a food waste context. For example, Parashar et al. (2020) present a study into supply chain enablers for carbon footprint reduction where they use the matrix of cross-impact multiplications applied to classification (MICMAC) analysis to categorize fourteen enablers (eg logistics, information sharing, traceability, technology, collaboration, regulation) as driving power and dependence power. Their research engaged three academic experts and two industrial practitioners (raw materials/ingredients suppliers to the hospitality sector).

As can be seen from the table 3, there is limited research into food waste analysis in the catering sector using MCDM, mainly the AHP/DEMATEL - base method. In this research, the DEMATEL method aims at revealing two groups of enablers: cause-enablers and effect-enablers. Efforts of decision makers should focus on cause-enablers that might eliminate, accordingly, enablers in the effect group. However, several enablers might be grouped as causes and decision makers hereby need to prioritize their efforts. The AHP method can tackle this prioritizing decision-making problem by quantifying the relative importance of cause-enablers in which cause-enablers with a high weight should be targeted before cause-enablers with low weight. Further features of the AHP and DEMATEL methods are presented in section 3.1 and 3.2, respectively.

Reference	MCDM	Description
Hung et al. (2006)	AHP	Find the optimal food waste management scheme.
Chen et al. (2014)	Entropy method and AHP	Assess the safety of animal feed food waste.
Babalola (2015)	AHP	Evaluate different waste management options and their applicability in Japan and select the most suitable waste treatment option.
Oprea & Gaceu (2016)	MCDM method	Identify the most appropriate waste and by-products suitable to be used in the bakery industry.
San Martin et al. (2017)	AHP and GIS	Evaluate the main parameters involved in the decision process and helps decision- makers to implement food waste valorisation strategies.
Chauhan et al. (2018)	ISM method	Identify and model the drivers of agri-food waste management in India.
Liu et al. (2018)	DEMATEL, ANP and VIKOR	Improve and select the location for the best food waste composting facilities.
Raut et al. (2018)	AHP	Identify the crucial causal factors of post-harvest losses in the fruits and vegetables supply chain in India
Parashar et al. (2020)	MICMAC technique	Identify the critical enablers for food supply chain management which impact the carbon footprint creation and develops a contextual relationship amongst the identified enablers

Table 3. MCDM methods used in food waste management.

#### 3. Research methodology

Figure 2 presents an overview of the methodology used in the research. The questionnaire to collect data for AHP and DEMATEL methods was designed to allow businesses to undertake a pairwise comparison concerned with the importance and causal relationship among the causes. The questionnaire was divided into three part: 1) general information about the business; 2) the company was asked to undertake a comparison between different food waste causes; 3) evaluation of the influence level for food waste causes.



Figure 2. Methodology overview.

We investigated the food waste enablers in this sector at five different levels (Figure 3) and they covered the major food management steps. Figure 3 presents the framework of food waste enablers based on the literature discussed in Section 2 and they are clustered into five levels: procurement and storage (A), processing (B), operation strategy (C), people (D) and consumer (E). As part of the methodology, the enablers are classified as a cause or an effect enabler and this grouping will help managers to identify of food waste and target them rather than randomly focus efforts on addressing all enablers.

The relative importance of each enabler in each level was quantified using AHP. Then, DEMATEL was applied to investigate the interrelationships among enablers and present them as causes and effects in a holistic diagram. Compared to the AHP method (Saaty, 1980), this approach considers the independencies among dimensions and enablers in addition to obtaining the importance degree of enablers. Using this approach, cause enablers have a strong impact on effect enablers and this interrelationship should be taken into account. According to the interdependence, the enablers could be divided into two groups: cause and effect, where enablers in the cause group should be targeted and solved before enablers in the effect group. The integrated priority ranks are generated as the cause enablers with high AHP weight before causes with low AHP weight; preceding to effect group with high AHP weight prior to effects with low AHP weight, to allow managers to identify the most conclusive determinants.



Figure 3. A framework of food waste enablers in catering industry.

The data for the AHP-DEMATEL method was collected using questionnaires that were distributed to forty companies with a response rate of 80%. Thirty-two responses consisted of 7 coffee shops, 5 takeaways, 13 restaurants and 7 pubs. The companies were randomly selected in one city and employees with management responsibilities filled in questionnaires. The number and diversity of participants allowed for distinctions among food waste enablers in addition to cause-effect analysis at different levels. Bloom et al. (2003) discuss that the most important factor in the amount of data collected for analysis to be sufficient is when there is enough data to identify distinctions. The authors also argue that the "more may not be better" as Rodgers et al. (1992) state that after identifying distinctions, each extra response increases the validity coefficient by almost 0.04. Therefore, considering the complexity of getting responses from, mainly a managerial level we have used the responses to proceed with this research.

# 3.1 AHP

Analytic Hierarchy Process (AHP) method was originally proposed and developed by Satty (1980) to support MCDM (De Felice and Petrillo 2014). This method quantifies the qualitative preferences of decision-makers and transfers them into measurable aspects (Ortiz et al. 2016). The AHP method analyses the sophisticated decisionmaking problem by conducting a number of pairwise comparisons among involved criteria and accordingly derive the criterion's weight. Since it was developed, AHP has proved its applicability for solving real-life decisionmaking problems. Govindan et al. (2015) and Chai et al. (2013) review research that identified AHP as one of the most successfully employed MCDM method due to its simplicity and ability to handle qualitative and quantitative criteria. Fallahpour & Moghassem (2012) discuss that AHP is one of the most popular MCDM methods in weighting problems. Due to simplicity, the method satisfies the requirement of current research in weighting food waste causes, AHP was also used to support managers in determining the priority order of these causes.

In this work, AHP was applied to determine the importance of weight for each ABCDE criteria and sub-criteria and Table 4 shows the evaluation scale of linguistic variables that were used to perform pairwise comparisons. Decision-makers need to give their opinion regarding the importance of each criteria / sub-criteria with respect to the others. AHP steps toward the solution also can be found in Satty (1980), Handfield et al. (2002), Mathiyazhagan et al. (2015), Gandhi et al. (2016).

The general AHP procedure consists of following steps (Satty, 1980; Gandhi et al., 2016):

9

*Step 1: Defining the causes:* The main food waste causes were identified based on knowledge and previous studies. *Step 2: Development of judgement matrix.* The participants were asked to evaluate relative importance using the scale developed by Satty (1980) (Table 4). For instance, if the decision-makers think that A1 is 'very strongly important' compared to A2 then we put 7 based on the evaluation scale.

Scores	Linguistic meaning
1	Equally Important (EQI)
3	Weakly Important (WI)
5	Strongly Important (SI)
7	Very Strongly Important (SI)

Extremely Important (EXI)

Table 4. AHP scores.

The judgement matrix A (eq.1) was calculated using the geometric average of all the comparison matrixes  $A_1, A_2, ..., A_m$ , where *m* and *n* denote the number of participants and criterion, respectively.

Please note: 2, 4, 6, 8 are the intermediate values

$$A = \sqrt[m]{A_1 \cdot A_2 \dots A_m} = \begin{bmatrix} 1 & a_{12} & \dots & a_{1i} & \dots & a_{1n} \\ a_{21} & 1 & \dots & a_{2i} & \dots & a_{2n} \\ \dots & \dots & \dots & \dots & \dots & \dots \\ a_{i1} & \dots & \dots & \dots & \dots & \dots \\ a_{n1} & a_{n2} & \dots & a_{ni} & \dots & 1 \end{bmatrix} \left( i, j = 1, 2, \dots n; \ a_{ij} = \frac{1}{a_{ji}} > 0 \right)$$
(1)

Step 3: Developing a normalised decision matrix. The matrix can be constructed by summing the column number in judgement matrix A and then dividing each value in the matrix by its column sum. Thus, column sum  $(S_j)$  and normalised decision matrix (B) (eq. 2, eq. 3):

$$S_i = a_{1i} + a_{2i} + \dots + a_{ni} \ (i = 1, 2, \dots, n)$$
(2)

$$B = \begin{bmatrix} \frac{1}{S_1} & \frac{a_{12}}{S_2} & \cdots & \frac{a_{1i}}{S_i} & \cdots & \frac{a_{1n}}{S_n} \\ \frac{a_{21}}{S_1} & \frac{1}{S_2} & \cdots & \frac{a_{2i}}{S_i} & \cdots & \frac{a_{2n}}{S_n} \\ \frac{a_{11}}{S_1} & \cdots & \cdots & \cdots & \cdots & \frac{a_{in}}{S_n} \\ \frac{a_{n1}}{S_1} & \frac{a_{n2}}{S_2} & \cdots & \frac{a_{ni}}{S_i} & \cdots & \frac{1}{S_n} \end{bmatrix}$$
(3)

Step 4: Determining the weight of each enabler: Calculate the average value of each row in the normalised matrix, the results for each row should be the weight  $(w_i)$  of the corresponding criteria (eq. 4).

$$w_i = \frac{1}{n} \left( \frac{a_{i1}}{S_1} + \frac{a_{i2}}{S_2} + \dots + \frac{a_{in}}{S_n} \right) \ (i = 1, 2, \dots, n) \tag{4}$$

Noticeably, the weight value for each criterion can constitute the eigenvector (W), which is

$$W = (w_1, w_2, \dots, w_i)^T (i = 1, 2, \dots, n)$$
(5)

*Step 5: Checking the consistency*. The consistency of the judgement matrix calculated using the Consistency Ratio (CR) that is equal to Consistency Index (CI) divided by Random Consistency Index (RCI) (eq. 6).

$$CR = \frac{CI}{RCI}$$
(6)

The CI value is related to the eigenvector (*W*), maximum eigen value ( $\lambda_m$ ) and order(*n*) of the judgement matrix (A), eq. 7.

$$CI = \frac{\lambda_m - n}{n - 1} \tag{7}$$

The maximum eigen value  $(\lambda_m)$  can be generated from the judgement matrix (A) and eigenvector (W) (eq. 8).

$$\lambda_m = \frac{1}{n} \cdot \sum_{i=1}^n \frac{(AW)_i}{w_i} \tag{8}$$

RCI value depends on the number of the judgements for a particular matrix of order n and presented in Table 5 (Satty, 1980).

Table 5. RCI value.

n	1	2	3	4	5	6	7	8	9	10
RCI	0	0	0.58	0.9	1.12	1.24	1.32	1.41	1.45	1.49

If the value of CR < 0.1, the judgement matrix could be considered at a rational consistency level. Otherwise, the judgement matrix might be inconsistent among some comparison values and needs further adjustment.

#### 3.2 Identifying causes and effects: DEMATEL

Decision Making Trial and Evaluation Laboratory (DEMATEL), which was first proposed by the United States Bastille laboratory in 1971, is extensively applied to analyse and evaluate the influence among all the criteria (Ortíz et al. 2016). This technique synthesizes the opinions or experience of experts by quantifying the influence level between each criterion. This method helps in identifying the relationship among involved criteria by categorizing them as either cause or effect. In other words, it reveals the correlation among criteria and can direct efforts of decision-makers in targeting causes rather than effects, which will be addressed accordingly (Büyüközkan & Çifçi, 2012; Mohammed, 2020). DEMATEL is described as a method for identifying causes-effects outcome in decision-making (Doraid et al., 2011; Nujoom et al., 2019; and Si et al., 2018). DEMATEL allows verification of interdependence among all selected criteria and development of a holistic cause-effect diagram. The implementation of the method has the following five steps (Tzeng et al. 2007, Wu and Tsai 2012; Yazdani et al. 2017):

*Step 1: Generating the direct-relation matrix from the questionnaire response.* Using the questionnaires, the participants were asked to score the influence level between each pair of criteria using scores presented in the Table 6. For instance, if a participant thinks that A1 has 'low influence' on A2, we would put score of '1'.

Table 6. Influence level scores in DEMAT	ce level scores in DEMATE	EL.
--	---------------------------	-----

Scores	Linguistic meaning
0	No influence
1	Low influence
2	Medium influence
3	High influence
4	Extreme influence

For each questionnaire, the response is converted into an influence matrix and the direct-relation matrix (C):

$$C = \frac{C_1 + C_2 + \dots + C_m}{m} = \begin{bmatrix} 0 & c_{12} & \dots & c_{1i} & \dots & c_{1n} \\ c_{21} & 0 & \dots & c_{2i} & \dots & c_{2n} \\ \dots & \dots & \dots & \dots & \dots & \dots \\ c_{i1} & \dots & \dots & \dots & \dots & \dots \\ c_{n1} & c_{n2} & \dots & c_{ni} & \dots & 0 \end{bmatrix}$$
(9)

Step 2: Computing the normalized direct-relation matrix (eq. 10, eq. 11.)

$$\mathbf{N} = x \cdot \mathbf{C} \tag{10}$$

where

$$x = \min\left[\frac{1}{\max\sum_{i=1}^{n} c_{ij}}, \frac{1}{\max\sum_{j=1}^{n} c_{ij}}\right] (i, j = 1, 2, ..., n)$$
(11)

*Step 3: Generating the total-relation matrix.* The total-relation matrix (T) (eq.12) reveals the total relationship including direct influence and indirect influence between each pair of criteria.

$$T = N(I - N)^{-1}$$
(12)

where I refers to the identical matrix.

Step 4: Dividing the criteria into cause and effect groups. Values  $D_i + R_j$  ("prominence") and  $D_i - R_j$  ("relation") were computed for each criterion.  $D_i$  and  $R_j$  are the sum of rows and columns of (T) (eq. 13 and 14).

$$D_{i} = \sum_{j=1}^{n} t_{ij} (i = 1, 2, ..., n)$$
(13)  
$$R_{j} = \sum_{i=1}^{n} t_{ij} (j = 1, 2, ..., n)$$
(14)

Where the "relation" value (i.e.  $D_i - R_j$ ) of one criterion is positive, this criterion would be regarded as a cause group. Where the "relation" value is negative, this criterion would be classified as an effect group.

Step 5: Developing a cause and effect diagram according to the threshold value. The setting of the threshold value ( $\alpha$ ) is helpful for a decision-maker to develop the cause and effect diagram (eq. 15). For example, when the value of total relationship between criterion *i* and criterion *j* in the matrix (T) is greater than the threshold value (i.e.  $t_{ij} > \alpha$ ), it means that criterion j is caused by criterion i, where the arrow begins from criterion *i* to criterion *j* and vice versa.

$$\alpha = \frac{1}{num} \sum_{i=1}^{n} \sum_{j=1}^{n} [t_{ij}]$$
(15)

where num is the number of elements in T

#### 4. Results

Different types of catering outlets have diverse customer profiles (demographics and behaviour) with different offerings where food contribution to total sales depends on many factors: location, menu, standards, management and service (Negus, 2004). By taking into account different types of business, the analysis is undertaken for all outlets in the catering industry under investigation and for each type to support the cross-comparison discussion in Section 5.

#### 4.1 Catering industry: all outlets

The AHP and DEMATEL analysis using 32 responses from all businesses presented in Table 7. The CR values are below 0.1 showing the high consistency across all responses. As shown in Table 7, the procurement and storage, and operation strategy have the highest AHP weights, indicating relatively higher importance towards food waste. Difficulty in forecasting (A<sub>1</sub>), strict regulations (B<sub>1</sub>) large menus (C<sub>3</sub>), miscommunication (D<sub>2</sub>) and over-ordering (E<sub>5</sub>) are the most important enablers in each level when considering only AHP weights, with the weight of 0.316, 0.369, 0.224, 0.278 and 0.335 respectively.

After obtaining enablers weights, DEMATEL method was used to categorize the food waste enablers into cause and effect and to develop a holistic cause-effect diagram. The category for each factor is derived using prominence value (D+R) and relation value (D-R) (see Table 7). At procurement and storage and processing levels, two enablers might lead to the three enablers leading to food waste. At the sub-level, difficulty in forecasting (A<sub>1</sub>) and customer attitude towards food (E<sub>1</sub>) are among enablers that have a high prominence and relation values among the sub- enablers.

Level enablers	AHP weight	D+R	D-R	Group	Priority Rank	Sub-level enablers	AHP weight	D+R	D-R	Group	Priority Rank
						A1	0.316	44.233	0.372	Cause	1
A. Dro ourom ont	0.256	36.8	0.69	Causa	1	A <sub>2</sub>	0.196	42.352	0.263	Cause	2
and Storage	0.230	04	0.08	Cause	1	A <sub>3</sub>	0.242	43.312	-0.052	Effect	4
and Storage						$A_4$	0.246	42.539	$\begin{array}{c c c c c c c c c c c c c c c c c c c $	3	
						B1	0.369	19.454	-0.105	Effect	3
D. Drossaring	0.17	37.3 88	0.87	Cause	2	<b>B</b> <sub>2</sub>	0.24	19.820	-0.779	Effect	4
D. Processing	0.17		7		2	<b>B</b> <sub>3</sub>	0.268	19.610	0.444	Cause	1
						$B_4$	0.123	17.111	0.440	Cause	2
C. Operation				Effect		C1	0.222	18.959	0.044	Cause	1
		38.1	- 0.71			$C_2$	0.21	18.540	-0.043	Effect	4
	0.237				3	C <sub>3</sub>	0.224	18.078	-0.456	Effect	3
sualegy		01	1			$C_4$	0.206	19.618	-0.275	Effect	5
						C <sub>5</sub>	0.138	16.227	0.731	Cause	2
						D1	0.192	61.166	-0.639	Effect	4
D. Decente	0.192	38.2	-	Effect	4	D <sub>2</sub>	0.278	60.129	-0.082	Effect	3
D. People	0.185	41	0.14	Effect	4	D <sub>3</sub>	0.263	61.395	0.229	Cause	2
			0			$D_4$	0.268	58.385	0.491	Cause	1
						E <sub>1</sub>	0.174	25.938	0.914	Cause	2
		25.0				E <sub>2</sub>	0.153	24.544	-0.320	Effect	5
E. Consumer	0.154	35.8 15	-0.7	Effect	5	E <sub>3</sub>	0.175	25.532	0.538	Cause	1
		15		Lileet		$E_4$	0.163	25.214	-0.509	Effect	4
						E <sub>5</sub>	0.335	25.040	-0.623	Effect	3

Table 7. AHP weights, priority ranking, cause and effect for each enabler across all outlets.

According to Yazdani et al. (2017), the elements in the cause group should be regarded as the main drivers to the elements in the effect group and can be considered as a priority. Using AHP results, the food waste enablers can be ranked following the principles that cause group (high AHP weight to low AHP weight) ranked before the effect group (high AHP weight to low AHP weight). The ranking guides the decision-makers to identify the most

significant food waste enablers to be targeted as a priority, for example in the procurement and storage. Difficulty in forecasting (A<sub>1</sub>), cooking amount unawareness (B<sub>3</sub>), inappropriate portion size (C<sub>1</sub>), attitude towards food waste (D<sub>4</sub>) and different dietary preference (E<sub>3</sub>) are the priority sub-enablers in each dimension.

Figure 4 depicts the holistic cause-effect diagram that reveals the contextual relationship and provides a visualisation of interrelationships among the food waste enablers. The value in brackets following the factor is the AHP weights and the number on the arrows specifies by how much one factor influences another. Processing is the only dimension that would affect the other four dimensions, especially the operation strategy and people, with a high influence level of 4.003 and 3.971. When the number of flow-out arrows is larger than the number of flow-in arrows, this enabler tends to affect others more than being influenced by others and deserve the attention. However, for some enablers, the numbers of flow-in and flow-out arrows are equal, therefore the influence level on the arrow could determine this enabler as cause or effect. Noticeably, lack of equipment and tools ( $B_4$ ), poor facility layout ( $C_5$ ) and attitude towards food waste ( $D_4$ ) are the enablers that affect preparation residue ( $B_2$ ), poor waste management ( $C_4$ ) and staff mishandling ( $D_1$ ) respectively without being influence levels in the same dimension which might imply the strong influence relationship of two enablers. For example, unskilled staff ( $D_3$ ) would significantly affect the  $D_1$  at the level of 7.961, while the threshold value and ordinary influence level are around 7.6 to 7.7. This might suggest that  $D_3$  will lead to  $D_1$ .



Figure 4. Cause and effect diagram for food waste enablers across all outlets in catering industry.

#### 4.2 Coffee shops

AHP weights for each food waste enabler in coffee shops have been calculated using responses from seven business (Table 8). The procurement and storage have the highest weight of 0.23; with processing ranked as the  $2^{nd}$  highest factor leading to food waste in the coffee shop. Among sub-level enablers, difficulty in forecasting (A<sub>1</sub>), cooking amount unawareness (B<sub>3</sub>), poor waste management (C<sub>5</sub>), unskilled staff (D<sub>3</sub>) and different dietary preference (E<sub>3</sub>) are the most crucial causes based on combined priority rankingwith weights of 0.352, 0.215, 0.149, 0.156 and 0.345. Enablers A, B and C are classified as causes. In A, difficulty in forecasting (A<sub>1</sub>) is the most significant enabler with highest prominence and relation value.

Level enablers	AHP weigh t	D+R	D-R	Group	Priorit y Rank	Sub- level enable rs	AHP weight	D+R	D-R	Group	Priorit y Rank
						A <sub>1</sub>	0.352	16.091	0.579	Cause	1
A. Dro gurger on t	0.22	15.007	0.427	Causa	1	$A_2$	0.26	15.149	-0.072	Effect	3
and Storage	0.23	15.097	0.437	Cause	1	A <sub>3</sub>	0.247	15.618	-0.727	Effect	4
and Storage						$A_4$	0.141	13.691	D-R         Group         Pr y 1           0.579         Cause           -0.072         Effect           -0.727         Effect           -0.72         Effect           -0.14         Effect           -0.529         Effect           -0.509         Effect           -0.057         Effect           -0.057         Effect           -0.572         Cause           -0.011         Effect           -0.732         Effect           -0.732         Effect           -0.483         Effect           -0.002         Effect           -0.002         Effect           -0.102         Effect           -0.102         Effect           -0.104         Effect	2	
						B1	0.409	14.319	-0.14	GroupJCauseEffectEffectCauseEffect	2
В.	0.10	15.448	0.625	Cause	2	$B_2$	0.204	13.909	-0.529	Effect	3
Processing	0.19					<b>B</b> <sub>3</sub>	0.215	14.698	0.751	Cause	1
						$B_4$	0.172	12.754	-0.083	Effect	4
						C1	0.285	18.046	-0.509	Effect	3
C Operation						C <sub>2</sub>	0.171	17.009	-0.057	Effect	5
C. Operation	0.182	15.532	0.535	Cause	3	C <sub>3</sub>	0.254	17.721	-0.82	Effect	4
suategy						$C_4$	0.141	19.046	0.572	Cause	2
						C5	0.149	18.515	0.814	Cause	1
						D1	0.266	42.229	-0.011	CauseEffectEffectEffectEffectEffectEffectEffectEffectEffectEffectCauseEffect	3
D. Poorla	0.205	16 575	0.867	Effort	4	D <sub>2</sub>	0.398	42.506	-0.732	Effect	2
D. People	0.203	10.373	-0.807	Effect	4	D <sub>3</sub>	0.156	39.555	1.226	Cause	1
						$D_4$	0.18	42.237	-0.483	Effect	4
						E <sub>1</sub>	0.202	18.123	-0.002	Effect	2
Б						E <sub>2</sub>	0.188	17.813	-0.102	Effect	3
E. Consumer	0.194	16.446	-0.73	Effect	5	E <sub>3</sub>	0.345	19.380	0.299	Cause	1
Consumer						$E_4$	0.121	17.814	-0.104	Effect	5
						E <sub>5</sub>	0.143	17.191	-0.091	1.727         Effect           0.22         Cause           0.14         Effect           0.529         Effect           0.529         Effect           0.529         Effect           0.529         Effect           0.509         Effect           0.083         Effect           0.057         Effect           0.529         Effect           0.509         Effect           0.509         Effect           0.572         Cause           .814         Cause           0.011         Effect           .732         Effect           .732         Effect           .026         Cause           .483         Effect           .002         Effect           .002         Effect           .102         Effect           .102         Effect           .104         Effect           .001         Effect	4

Table 8. AHP weights, priority ranking, cause and effect for each enabler in coffee shops.

Figure 5 presents the holistic cause-effect diagram of the food waste enablers in coffee shops. Procurement and storage, processing and operation strategy are the dimensions that would influence people and consumer, without being affected by others. These two affected dimensions would interact with each other at a similar influence level. Noticeably, among the sub-enablers, unskilled staff ( $D_3$ ) could be regarded as the root cause in its dimension because it would influence the other sub-enablers without being impacted.



Figure 5. Cause and effect diagram for food waste enablers in coffee shops.

# 4.3 Takeaways

Table 9 presents the analysis and findings related to takeaways. The procurement and storage (A) is the most important contributor towards food waste at around 47% of the total weight. Difficulty in forecasting (A<sub>1</sub>) and strict regulation (B<sub>1</sub>) are effects. Procurement and storage (A) and consumer (E) enablers could be recognized as the causes leading to other enablers. For the sub-level, customer attitude towards food (E<sub>1</sub>) has the highest prominence and relation values. Regarding the priority ranking (see Table 9), procurement and storage (A) is the priority level. Lack of purchase plan (A<sub>2</sub>), cooking amount unawareness (B<sub>3</sub>), poor facility layout (C<sub>5</sub>), attitude towards food waste (D<sub>4</sub>) and E<sub>1</sub> are the priority for each level.

Level enablers	AHP weight	D+R	D-R	Group	Priorit y Rank	Sub-level enablers	AHP weight	D+R	D-R	Group	Priority Rank
						A1	0.428	23.887	-1.407	Effect	4
A.	0.465	22.042	1 (02	Course	1	A <sub>2</sub>	0.225	23.699	0.671	Cause	1
and Storage	0.465	23.042	1.003	Cause	1	A <sub>3</sub>	0.196	24.77	0.483	Cause	2
and Storage						$A_4$	0.152	22.688	0.252	Group Effect Cause Cause Effect Effect Cause Cause Cause Cause Cause Cause Cause Cause Effect	3
						B1	0.399	11.444	-0.569	Effect	3
P. Processing	0.112	25.214	-0.228	Effect	5	$B_2$	0.129	12.501	-0.611	Effect	4
D. Processing	0.112				3	<b>B</b> <sub>3</sub>	0.364	11.902	0.745	Cause	1
						$B_4$	0.108	10.435	0.435	Cause	2
		24.523	-0.56	Effect		C1	0.214	9.666	0.244	Cause	2
C. Operation strategy						$C_2$	0.208	8.897	0.004	Cause	3
	0.197				3	C <sub>3</sub>	0.127	7.862	0.268	Cause	4
						$C_4$	0.233	9.324	-1.158	Effect	5
						C <sub>5</sub>	0.219	8.473	0.642	Cause	1
						$D_1$	0.256	10.691	-0.871	Effect	3
D People	0.157	25.14	0.836	Effect	4	D <sub>2</sub>	0.188	9.731	0.634	Cause	2
D. reopie	0.137	23.14	-0.850	Effect	4	D <sub>3</sub>	0.251	10.037	-0.21	Effect	4
						$D_4$	0.305	9.909	0.447	Cause	1
						E <sub>1</sub>	0.236	9.45	0.92	Cause	1
						E <sub>2</sub>	0.149	8.547	-0.488	Effect	5
E. Consumer	0.07	24.329	0.021	Cause	2	E <sub>3</sub>	0.267	9.2	-0.145	Effect	3
						$E_4$	0.1	8.772	0.095	Cause	2
						E <sub>5</sub>	0.248	7.929	-0.382	Effect	4

Table 9. AHP weights, priority ranking, cause and effect in takeaways.

Figure 6 presents the cause and effect diagram for takeaways and it is more complicated compared to other businesses. Procurement and storage (A) is shown as the core dimension influencing other dimensions but not affected by others. Among the sub-enablers, lack of equipment and tools (B<sub>4</sub>) fully impacts on preparation residue (B<sub>2</sub>), miscommunication (D<sub>2</sub>) and attitude towards food waste (D<sub>4</sub>) would affect staff mishandling (D<sub>1</sub>) and unskilled staff (D<sub>3</sub>). As for the influence level, the numbers on the arrow from A<sub>3</sub> to A<sub>1</sub>, from B<sub>3</sub> to B<sub>2</sub> and from C<sub>1</sub> to C<sub>4</sub> are relatively higher in their level at 3.424, 1.864 and 1.212 respectively. This implies that forgotten and spoiled food (A<sub>3</sub>), cooking amount unawareness (B<sub>3</sub>), and inappropriate portion size (C<sub>1</sub>) would strongly affect the difficulty in forecasting (A<sub>1</sub>), preparation residue (B<sub>2</sub>) and poor waste management (C<sub>4</sub>).



Figure 6. Cause and effect diagram for food waste enablers in takeaways.

#### 4.4 Restaurants

Table 10 presents AHP/DEMATEL findings related to restaurants. Operation strategy has the highest AHP weight of 0.247. Over-stocking (A<sub>4</sub>), strict regulation (B<sub>1</sub>), large menu (C<sub>3</sub>), unskilled staff (D<sub>3</sub>) and over-ordering (E<sub>5</sub>) also have the highest AHP weights for each dimension respectively. The analysis of cause and effect groups shows that processing (B) and people (D) play an important cause on other enablers and there is only one sub-factor grouped as an effect (preparation residue (B<sub>2</sub>) and attitude towards food waste (D<sub>4</sub>)). Processing is the dimension with the first consideration in restaurants due to its high cause weight (0.195) and priority ranking of 1. Within the sub-enablers, difficulty in forecasting (A<sub>1</sub>), strict regulation (B<sub>1</sub>), inappropriate portion size (C<sub>1</sub>), unskilled staff (D<sub>3</sub>) and customer attitude towards food (E<sub>1</sub>) should be addressed first.

Level enablers	AHP weight	D+R	D-R	Group	Priority Rank	Sub – level enablers	AHP weight	D+R	D-R	Group	Priority Rank
						A <sub>1</sub>	0.233	33.352	1.386	Cause	1
A. Drogunger on t	0.217	20.252	0.442	Effect	4	A <sub>2</sub>	0.173	33.303	0.16	Cause	2
and Storage	0.217	20.235	-0.442	Effect	4	A <sub>3</sub>	0.294	33.394	-0.415	Effect	4
and Storage						$A_4$	0.3	33.835	-1.131	Group Pr Cause Cause Ca	3
						B <sub>1</sub>	0.306	10.824	0.054	Cause	1
В.	0.105	10.052	0.577	Causa	1	$B_2$	0.303	11.583	-0.685	Effect	4
Processing	0.195	19.933	0.577	Cause	1	B <sub>3</sub>	0.277	11.167	0.138	Cause	2
						$B_4$	0.114	9.077	0.494	Cause	3
				Effect		C1	0.208	15.48	0.248	Cause	1
C. Operation strategy		20.794	-0.879			C <sub>2</sub>	0.194	15.866	0.177	Cause	2
	0.247				3	C <sub>3</sub>	0.245	14.811	-0.213	Effect	4
						C4	0.235	16.298	-0.307	Effect	5
						C <sub>5</sub>	0.118	12.259	0.096	Cause	3
					2	D1	0.147	41.936	0.425	Cause	3
D. Poorla	0.182	20.067	1 244	Causa		D <sub>2</sub>	0.256	41.856	0.318	Cause	2
D. reopie	0.162	20.007	1.344	Cause		D3	0.326	42.419	0.376	Cause	1
						$D_4$	0.27	39.012	-1.119	Effect	4
						E <sub>1</sub>	0.155	20.954	0.58	Cause	1
Б						E <sub>2</sub>	0.117	19.836	-0.151	Effect	5
E. Consumer	0.159	19.312	-0.6	Effect	5	E <sub>3</sub>	0.114	19.963	0.685	Cause	2
Consumer						$E_4$	0.196	20.809	-0.79	Effect	4
						E <sub>5</sub>	0.419	21.971	-0.323	Effect	3

Table 10. AHP weights, priority ranking, cause and effect in restaurants.

As shown in Figure 7, processing and people are root enablers that impact on others. However, it can be argued that decision-makers should target the people dimension prior to processing because people would affect processing while processing would not affect people. Hence, a slight discrepancy between this argument and the combined priority rank. This difference may be attributed to the higher AHP weight of processing than people, where processing is considered first when they both belong to a cause group. For sub-level enablers, difficulty in forecasting (A<sub>1</sub>), lack of equipment and tools (B<sub>4</sub>) and different dietary preference (E<sub>3</sub>) are the causes that entirely affect others without being impacted. The high influence levels of cooking amount unawareness (B<sub>3</sub>) on preparation residue (B<sub>2</sub>) and A<sub>1</sub> on over-stocking (A<sub>4</sub>) demonstrate the powerful causality of those enablers. Noticeably, the poor facility layout (C<sub>5</sub>) neither has an impact on other enablers nor is influenced by others while it has been grouped as a cause. A possible explanation for this might be that C<sub>5</sub> impacts on others more than being impacted even though its overall relationship with other enablers is quite weak and would not be shown in the diagram.



Figure 7. The cause and effect diagram for food waste enablers in restaurants.

# 4.5 Pubs

Table 11 presents the AHP/DEMATEL results related to pubs. Operation strategy (C) has the highest weight of 0.282 and it indicates relatively high importance towards food waste. For each sub-level, over-stocking (A<sub>4</sub>), strict regulation (B<sub>1</sub>), buffet-style (C<sub>2</sub>), attitude towards to food waste (D<sub>4</sub>), and over-ordering (E<sub>5</sub>) have highest AHP weights, especially B<sub>1</sub> and E<sub>5</sub> with weight values of 0.405 and 0.481. Furthermore, combined priority ranking identifies the procurement and storage level as the most contributing factor towards food waste and should be addressed first and within sub-enablers, A<sub>3</sub>, B<sub>1</sub>, C4, D<sub>4</sub> and E<sub>1</sub> are priorities. Cause and effect groups illustrate that procurement and storage (A), processing (B) and consumer (E) are the causes and there is only one cause in the people dimension - attitude towards food waste (D<sub>4</sub>).

Level enablers	AHP weight	D+R	D-R	Group	Priority Rank	Sub – level enablers	AHP weight	D+R	D-R	Group	Priority Rank
						$A_1$	0.331	18.062	-0.511	Effect	4
A. Proguramont	0.221	16.4	0.67	Cause	1	A <sub>2</sub>	0.144	15.918	0.084	Cause	2
and Storage	0.231	63	0.67			A <sub>3</sub>	0.165	16.596	0.788	Cause	1
and Storage						$A_4$	0.36	17.685	-0.361	Effect	3
	0.138			Cause	3	$B_1$	0.405	21.456	0.157	Cause	1
В.		16.5	0.24			$B_2$	0.259	19.762	0.15	Cause	2
Processing		22	1			<b>B</b> <sub>3</sub>	0.231	19.844	-0.523	Effect	4
						$B_4$	0.105	19.446	0.216	Cause	3
					4	C1	0.18	13.87	-0.173	Effect	5
C. On which		17.0	-	Effect		$C_2$	0.276	13.388	-0.452	Effect	3
c. Operation strategy	0.282	17.2	0.42 1			C <sub>3</sub>	0.237	14.23	-0.843	Effect	4
		00				$C_4$	0.194	14.234	0.445	Cause	1
						C <sub>5</sub>	0.113	10.487	1.024	Cause	2

Table 11. AHP weights, priority ranking, cause and effect in pubs.

	0.164			Effect	5	$D_1$	0.16	12.388	-0.109	Effect	4
D. People		16.7	0.81 9			$D_2$	0.264	12.153	-0.644	Effect	3
		75				$D_3$	0.267	13.773	-0.26	Effect	2
						$D_4$	0.309	11.939	1.014	Cause	1
	0.185			Cause	2	E <sub>1</sub>	0.103	18.062	0.709	Cause	1
F		12.0	0.22			E <sub>2</sub>	0.157	16.89	-0.042	Effect	4
E. Consumer		13.0	0.32 8			E <sub>3</sub>	0.103	17.292	0.564	Cause	2
Consumer		01				$E_4$	0.156	17.186	-0.382	Effect	5
						E <sub>5</sub>	0.481	16.504	-0.849	Effect	3

Figure 8 presents the relationships between enablers and sub-enablers. It is worthy to note that the consumer dimension neither affects other dimensions nor is impacted by others. Procurement and storage influence relatively the operation strategy (1.961) in pubs. As for the sub-enablers, forgotten and spoiled food ( $A_3$ ) and poor facility layout ( $C_5$ ) are the cause enablers that only affect others, while over-ordering ( $E_5$ ) is the effect factor that is only impacted by others. The influence level of strict regulation ( $B_1$ ) on cooking amount unawareness ( $B_3$ ) is quite high compared to others (2.814).



Figure 8. The cause and effect diagram for food waste enablers in pubs.

#### 5. Discussion

Tables 12 and 13 present a summary of the findings across all businesses (G) in the catering industry and for individual outlets to allow a cross-business comparison for coffee shops (CS), takeaways (T), restaurants (R) and pubs (P). The table presents AHP weights, cause or effect classification and combined priority ranking. The values

highlighted in both tables support discussion presented in this section and emphasize results that will help decision-makers to prioritize their improvement efforts, by targeting the most influential levels first.

*Cross-comparison among businesses:* AHP weights represent the business opinions towards the importance of each food waste enabler. When considering all businesses (G), the procurement and storage level revealed the highest weigh (0.256) followed by operation strategy (C) (0.237), people (D) (0.183), processing (B) (0.17) and consumer (E) (0.154). The results indicate that food waste causes in the procurement and storage level play a paramount role and should be addressed by decision-makers as a priority. At sub-level analysis, coffee shops and takeaways have similar views for the difficulty in forecasting (A<sub>1</sub>) enabler when considering AHP weights. One of the reasons for that grouping can be the size of businesses where the majority of coffee shops and takeaways are micro-companies while the size of restaurants and pubs can be bigger, which might suggest that similar business size models could have similar attitudes towards food waste. DEMATEL classifies food waste enablers as causes and effects where cause enablers should be targeted as a priority with urgent attention. It is worthy to notice that some enablers are causes in three food outlets but regarded as the effect in others. For example, lack of purchase plan (A<sub>2</sub>), lack of equipment and tools (B<sub>4</sub>) and customer attitude towards food (E<sub>1</sub>) are causes in takeaways, restaurants and pubs while it is an effect in coffee shops.

Land		Group					Priority rank								
Level	G	CS	Т	R	Р	G	CS	Т	R	Р	G	CS	Т	R	Р
A. Procurement and storage	0.256	0.23	0.465	0.217	0.231	С	С	С	Е	С	1	1	1	4	1
B. Processing	0.17	0.19	0.112	0.195	0.138	С	С	Е	С	С	2	2	5	1	3
C. Operation strategy	0.237	0.182	0.197	0.247	0.282	Е	С	Е	Е	Е	3	3	3	3	4
D. People	0.183	0.205	0.157	0.182	0.164	Е	Е	Е	С	Е	4	4	4	2	5
E. Consumer	0.154	0.194	0.07	0.159	0.185	Е	Е	С	Е	С	5	5	2	5	2

Table 12. Level enablers summary.

C-cause; E-effect

The combined priority rank demonstrates which food waste enablers should be regarded as a conclusive determinant and should be addressed first (Tables 12, 13). Procurement and storage level is the priority for majority except for restaurants where processing takes precedence. Furthermore, takeaways should focus on the lack of a purchase plan and pubs on forgotten and spoiled food. At the processing level cooking amount unawareness should be considered first in coffee shops, takeaways and strict regulation is the priority for restaurants and pubs.

Level	evel Sub-level		AHP weight					Group					Priority rank				
enablers	enablers	G	CS	Т	R	Р	G	CS	Т	R	Р	G	CS	Т	R	Р	
	Difficulty in forecasting (A <sub>1</sub> )	0.316	0.352	0.428	0.233	0.331	С	С	Е	С	Е	1	1	4	1	4	
	Lack of purchase plan $(A_2)$	0.196	0.260	0.225	0.173	0.144	С	Е	С	С	С	2	3	1	2	2	
А	Forgotten and spoiled food (A <sub>3</sub> )	0.242	0.247	0.196	0.294	0.165	Е	Е	С	Е	С	4	4	2	4	1	
	Over-stocking (A <sub>4</sub> )	0.246	0.141	0.152	0.300	0.360	Е	С	С	Е	Е	3	2	3	3	3	
	Strict regulation on safety and quality (B <sub>1</sub> )	0.369	0.409	0.399	0.306	0.405	Е	Е	Е	С	С	3	2	3	1	1	
л	Preparation residue (B <sub>2</sub> )	0.240	0.204	0.129	0.303	0.259	Е	Е	Е	Е	С	4	3	4	4	2	
D	Unaware of cooking amount (B <sub>3</sub> )	0.268	0.215	0.364	0.277	0.231	С	С	С	С	Е	1	1	1	2	4	
	Lack of equipment and tools (B <sub>4</sub> )	0.123	0.172	0.108	0.114	0.105	С	Е	С	С	С	2	4	2	3	3	
	Inappropriate portion size (C <sub>1</sub> )	0.222	0.285	0.214	0.208	0.180	С	Е	С	С	Е	1	3	2	1	5	
	Buffet style leading to over- taking (C <sub>2</sub> )	0.210	0.171	0.208	0.194	0.276	Е	Е	С	С	Е	4	5	3	2	3	
С	Large menus (C <sub>3</sub> )	0.224	0.254	0.127	0.245	0.237	Е	Е	С	Е	Е	3	4	4	4	4	
	Poor waste management (C <sub>4</sub> )	0.206	0.141	0.233	0.235	0.194	Е	С	Е	Е	С	5	2	5	5	1	
	Poor layout of facility (C <sub>5</sub> )	0.138	0.149	0.219	0.118	0.113	С	С	С	С	С	2	1	1	3	2	
	Staff mishandling (D <sub>1</sub> )	0.192	0.266	0.256	0.147	0.160	Е	Е	Е	С	Е	4	3	3	3	4	
D	Miscommunication between management & staff (D <sub>2</sub> )	0.278	0.398	0.188	0.256	0.264	Е	Е	С	С	Е	3	2	2	2	3	
	Unskilled staff	0.263	0.156	0.251	0.326	0.267	С	С	Е	С	Е	2	1	4	1	2	
	Attitude towards food waste $(D_4)$	0.268	0.180	0.305	0.270	0.309	С	Е	С	Е	С	1	4	1	4	1	
	Customer attitude, values and behaviours towards food (E <sub>1</sub> )	0.174	0.202	0.236	0.155	0.103	С	Е	С	С	С	2	2	1	1	1	
	Random customer purchase $(E_2)$	0.153	0.188	0.149	0.117	0.157	Е	Е	Е	Е	Е	5	3	5	5	4	
Е	Different dietary preference of customers $(E_3)$	0.175	0.345	0.267	0.114	0.103	С	С	Е	С	С	1	1	3	2	2	
	Considering food unattractive (E <sub>4</sub> )	0.163	0.121	0.100	0.196	0.156	Е	Е	С	Е	Е	4	5	2	4	5	
	Over-ordering (E <sub>5</sub> )	0.335	0.143	0.248	0.419	0.481	Е	Е	Е	Е	Е	3	4	4	3	3	

T-1-1-	12	C., h 1,1		
Table	13.	Sub-level	enablers	summary.

C-cause; E-effect

*Procurement and storage:* Across the sector (G), in CS, T and P it has been discussed that the focus should be on procurement and storage and this finding is consistent with Charlebois et al. (2015). However, in our study, restaurants regard procurement and storage as 4<sup>th</sup> priority (see column 14 for combined priority rank, in Table 12) but it does not mean that restaurants can ignore this factor because the AHP weight is the second highest for restaurants. The lower priority rank may be attributed to the interrelationship of five dimensions, where the procurement and storage factor in the restaurants is more susceptible to other cause enablers compared to other types of business. This suggests that restaurant managers should address this factor after other cause enablers.

Difficulty in forecasting (A<sub>1</sub>) is the conclusive determinant in G, CS and R. This enabler could lead to overstocking and forgotten and spoiled food (Charlebois et al., 2015; Derqui et al., 2016), which is also supported by our findings. The management implication is that CS and R could collect and analyse relevant historical data to improve their forecasting ability and prevent food waste effectively (Priefer et al., 2016). However, the findings from T and P illustrate the implication in a different direction. In takeaways, the determinant cause of food waste is lack of a purchase plan instead of difficulty in forecasting. The purchase plan is an important aspect that strongly impacts on food waste generation in the industry (Baldwin, 2012; Charlebois et al., 2015). This finding reveals that managers in takeaways should purchase the raw material with appropriate planning and organizing to reduce inventory surplus.

*Processing:* The findings support the literature that processing plays an important role in food waste generation (Derqui et al., 2016; Priefer et al., 2016). For restaurants, it has the highest rank and for G, CS and P, this level takes second and third priorities. Regarding sub-enablers, unawareness of cooking amount is the most contributing factor towards food waste in G, CS and T, while the strict regulation in restaurants and pubs. For most coffee shops and takeaways, the food (e.g. sandwiches, chips, fried chicken) is often pre-done before customers coming, but most restaurants and pubs tend to follow the "make to order" principle and confirm the exact cooking amount in advance, which is probably why unawareness of cooking amount has a higher impact on coffee shops and takeaways. Furthermore, unsuitable food due to preparation residue, over-cooked food and food waste due to lack of equipment and tools would be disposed of due to existing food safety regulations (Charlebois & Hielm, 2014). Therefore, the managers in restaurants and pubs should focus on (1) targeting the root causes of food waste and reduce them accordingly; and (2) establishing a treatment plan to handle waste (i.e., overproduction) that can be used as, for instance as pet food or compost (Betz et al., 2015).

*Operation strategy:* For G, CS, T and R, the operation strategy level is positioned as 3<sup>rd</sup> in the priority, whereas pubs consider this dimension as 4<sup>th</sup>. Papargyropoulou et al. (2016) discuss that restaurant operations are one of

the main food waste drivers, and our study indicates that operation strategy is important and should not be ignored and addressed in order of priorities. Inappropriate portion size seems to be the most critical factor for the sector (G) and restaurants, and it is consistent with Betz et al. (2015). This suggests that restaurants can deal with excessive size by adapting the portion size to customer needs (Sakaguchi et al., 2018). For CS and takeaways, poor facility layout is the main factor for food waste generation, and our findings are different to the study by Derqui et al. (2016) where kitchen design and layout in the restaurant was not an important contributor to food waste. Panisello & Quantick (2001) discuss that the poor facility layout may hinder communication or service of the food, leading to food waste. In pubs, poor waste management is the main waste factor; therefore designing effective waste disposal practice could be beneficial. Poor waste management is also seen as the least important factor across the sector (G), takeaways and restaurants. This result reveals that food waste management has not received enough attention.

*People:* This level mainly refers to the food waste enablers related to the staff and for G, CS and T, it is ranked as 4<sup>th</sup>, whereas it comes 5<sup>th</sup> for pubs. Restaurants have a different view where people dimension should be taken into account as a second priority. This difference may imply that the staff enablers have a stronger impact on the restaurants than other types, thus restaurants should be more concerned with staff enablers to minimise the food waste effectively. The literature also discusses that staff enablers contribute considerably towards the food waste in the catering industry (Baldwin, 2012; Charlebois et al., 2015).

Usually for the business, collecting and recycling food waste can be costlier compared to disposal, thus staff are allowed to throw away food waste into the general bins without any further action (Gustavsson et al. 2011). Our findings indicate that attitudes toward food waste are the main cause of waste when some food could have been reused rather than disposed of. Takeaways and pubs should raise staff awareness related to sustainability through staff training and appropriate actions. In CS and R, the unskilled staff is a major contributor to food waste where staff mishandling and miscommunication in the workplace should be improved through essential staff skills training.

*Consumer:* This level considers food waste enablers that come from customers. Across the sector (G), CS and R, it is identified as the last priority for consideration, where customers might not be crucial to minimise the food waste in the context of this study. Considering sub-enablers, different dietary preferences are a conclusive determinant across the sector (G) and for coffee shops. This could increase the difficulty in determining procurement, preparation and cooking amounts (Baldwin, 2012). Noticeably, different dietary preferences are

objective enabler because food companies cannot control what customers would like to eat. For coffee shops, this cause can be handled by communicating with customers regarding serving size before preparing and cooking. In the other three business types, customer attitudes, values and behaviours towards food are considered as the primary cause of food waste which was supported by Lorenz et al. (2017). Companies should create customer awareness related to waste issues through active engagement. An unattractive food factor has low priority in this study, and it could be linked to the fact that most businesses are unwilling to admit that their food is not attractive.

## 6. Conclusions

The research presented a food waste mitigation methodology that should help businesses to design effective food waste prevention and reduction strategies. To this end, this paper aims to identify the main enablers of food waste and then guide managers to target root causes by conducting the cause-effect analysis. A hybrid AHP-DEMATEL approach is proposed to perform this analysis and to identify the relative importance of cause and effect enablers. This paper targeted the catering sector in identifying five levels (i.e., procurement and storage, processing, operation strategy, people and consumer). Data required for applying the AHP-DEMATEL approach was collected from 32 participants with management responsibilities from businesses in the catering sector in Wales, UK. Based on an application of AHP-DEMATEL approach, a holistic cause-effect diagram was developed to graphically illustrate the relationships among causes and effects of food waste. Those groups were also prioritized and sequenced vis-à-vis their relative importance from practitioners' perspectives.

#### 6.1 Theoretical and managerial implications

This research provided an original contribution both to the academic research field and to practitioners. The proposed methodology used multi-criteria decision-making methods and provided new insights into major food waste enablers when considering the level of importance and causal relationship simultaneously. It presented a holistic cause-effect analysis for food waste enablers and evaluated and compared the importance of interrelationships and assessed the priorities among various food waste factors. This study, for example, shows that the procurement and storage is the cause of food waste for all outlets except for restaurants where this enabler is an effect. The same observation is for processing in the takeaway. Poor facility layout is the only cause across all outlets that indicate its strong impact on operation strategy. The cause-effect analysis played a key role in the identification of where to design future strategies to mitigate food waste for different types of businesses. It would not be possible to achieve without the use of MCDM methods. The AHP method hereby helped in revealing the most important food waste enablers; that might guide the strategies and priorities towards food waste mitigation.

Furthermore, the DEMATEL method allowed to perform cause-effect analysis for all food waste enablers and sub-enablers towards understanding the interrelationship among them and accordingly, categorize them as cause enablers and effect enablers. The identified cause enablers are prioritized based on the AHP outcome. Therefore, it can be argued that the research methodology presented in this paper can be re-visited by scholars in food waste management to explore root causes of food waste for different sectors and at different stages throughout the supply chain. In addition, it could also benefit researchers in sustainable supply chain management to undertake cause-effect and relative importance analysis of a particular topic's enablers. This methodology can also be used to categorize enablers of supply chain resilience as causes and effects and then prioritize targeted causes based on their weight. Similarly, this could improve the management and decision making of outsourcing risks.

From an industrial marketing perspective, the results will support the design of a more effective sustainable marketing strategy, continuous development and commitment to a greener corporate image. This methodology could also be used to analyse causes and effects of traditional criteria (e.g., cost, and quality) and green criteria (e.g., green resources, and low carbon footprint) to understand consumers' commitment towards greener products. The research contributions inspire B2C and B2B firms towards more dedicated food waste mitigation and reduction strategies in the downstream supply chain, instead of strategies that might lead to temporary enhancement as they direct both enablers and causes. Thus, arguably, this paper contributes to the causes-effects analysis literature, not merely in proposing the AHP-DEMATEL methodology, in targeting the food waste, but also in exploring how this methodology leads to an in-depth analysis of a multifaceted causes-effects industrial marketing or supply chain problems orchestrated by a holistic, instead of compendious, methodological-analytical tactic.

For practitioners, the proposed methodology could help food businesses to recognize the levels of importance and interrelationships of various enablers that exist in different dimensions. By understanding degrees of importance and interrelationships, the business could assess the priority of food waste enablers in each dimension. In doing so, companies can design effective strategies to reduce waste by targeting the most significant causes instead of dealing with all the causes at the same time. For example, the research outcome demonstrated that decision makers in all outlets except for restaurants would be encouraged to address food waste causes in the procurement and storage as the first priority as they revealed the most relative important causes. Decision makers should also notice that some enablers could be either causes or effects based on the considered sector/business type. In this work, for instance, they should target lack of purchase plan, lack of equipment and tools, and customer attitude towards food in takeaways, restaurants and pubs and ignore them in coffee shops as they represent as effects in the latter. In the same context, the results showed that for restaurants, processing has the highest importance, and for coffee shops and pubs it had second and third priorities. Also, the procurement and storage enabler turned out to represent one of the most effective factor in restaurants. It is well known that resources are scarce in this sector, as a large proportion of companies are considered as micro-businesses or SMEs. Therefore, addressing the most crucial factors of food waste could reduce food waste issues with limited resources and create the maximum value for the business. The developed methodology could be applied as a strategic tool in daily operations to prevent food waste in advance by managing and controlling the major causes.

## 6.2 Future research avenues

Future research can include different actors in the entire supply chain. The proposed food waste mitigation approach could be used to undertake cause-effect analysis for food waste enablers at a business-to-business level between, for example, suppliers and distributors, between suppliers and food manufacturers or between distributors and foodservice providers. This may help to identify further correlations and relationships among food waste enablers throughout the supply chain that can be addressed via a more integrated supply chain management. This approach will help to establish visibility throughout the supply chain towards food waste mitigation via shared food waste enablers. Difficulty in forecasting was identified as a cause enabler, thus it would be beneficial to undertake analysis of the relationship between current forecasting and inventory management approaches to understand links between food waste and supporting strategies. This could lead to the development of revised forecasting/inventory management techniques to mitigate food waste accordingly. On the people level, mishandling and unskilled staff were highlighted as an enabler of food wasted could be another area of exploration. Another research avenue could explore the behaviour of customers and managers in different cultures to allow further transferability of the findings presented.

### References

- Aschemann-Witzel, J., Giménez, A., Grønhøj, A. and Ares, G., 2020. Avoiding household food waste, one step at a time: The role of self-efficacy, convenience orientation, and the good provider identity in distinct situational contexts. Journal of Consumer Affairs, 54(2), pp.581-606.
- Babalola, M.A. (2015). A multi-criteria decision analysis of waste treatment options for food and biodegradable waste management in Japan. *Environments*, *2*, 471-488.

- Baldwin, C.J. (2012). Greening Food and Beverage Services: A Green Seal Guide to Transforming the Industry.Washington D.C.: Green Seal Inc.
- BCFN (2012). Food waste: causes, impacts and proposals. Barilla Centre for Food and Nutrition (BCFN). Available at: <u>https://barillacfn.com/m/publications/food-waste-causes-impact-proposals.pdf</u> [Accessed: 20 July 2018].
- Betz, A., Buchli, J., Göbel, C., & Müller, C. (2015). Food waste in the Swiss food service industry magnitude and potential for reduction. *Waste Management*, *35*, 218-226.
- Bharucha, J. (2018). Tackling the challenges of reducing and managing food waste in Mumbai restaurants. *British Food Journal*, *120* (*3*), 639-649.
- Bilska, B., Wrzosek, M., Kołożyn-Krajewska, D., & Krajewski, K. (2016). Risk of food losses and potential of food recovery for social purposes. *Waste Management*, 52, 269-277.
- Bloom, M., Fischer, J., & Orme, J. G. (2003). Evaluating practice: Guidelines for the accountable professional (4th ed.). Boston, MA: Allyn & Bacon.
- Bonadonna, A., Matozzo, A., Giachino, C. and Peira, G., 2019. Farmer behavior and perception regarding food waste and unsold food. British Food Journal.
- Brindley, C. & Oxborrow, L. (2014). Aligning the sustainable supply chain to green marketing needs: A case study. *Industrial Marketing Management*, 43 (1), 45-55.
- Büyüközkan, G., & Çifçi, G. (2012). A novel hybrid MCDM approach based on fuzzy DEMATEL, fuzzy ANP and fuzzy TOPSIS to evaluate green suppliers. *Expert Systems with Applications*, *39* (*3*), 3000–3011.
- Buzby, J.C., Hyman, J., Stewart, H., & Wells, H.F. (2011). The value of retail and consumer level fruit and vegetable losses in the United States. *Journal of Consumer Affairs*, 45(3), 492-515.
- Chai, J., Liu, J.N.K., & Ngai, E.W.T. (2013). Application of decision-making techniques in supplier selection: A systematic review of literature. *Expert Systems with Applications, 40 (10), 3872–3885*.

- Chan, H., He, H., & Wang, W. (2012). Green marketing and its impact on supply chain management in industrial markets. *Industrial Marketing Management*. 41 (4), 557–562.
- Charlebois, S., & Hielm, S. (2014). Empowering the regulators in the development of national performance measurements in food safety. *British Food Journal*, *116*(2), 317-336.
- Charlebois, S., Creedy, A., & von Massow, M. (2015). "Back of house" focused study on food waste in fine dining: The case of Delish restaurants. *International Journal of Culture, Tourism and Hospitality Research*, 9(3), 278-291.
- Chauhan, A., Debnath, R. M., Singh, S.P. (2018). Modelling the drivers for sustainable agri- food waste management. *Benchmarking: An International Journal*, 25(3), 981-993.
- Chen, T., Yiying, J., Qiu, X., & Chen, X. (2014). A hybrid fuzzy evaluation method for safety assessment of foodwaste feed based on entropy and the analytic hierarchy process methods. *Expert Systems With Applications*, 41(16), 7328-7337.
- Cicatiello, C., Franco, S., Pancino, B., & Blasi, E. (2016). The value of food waste: An exploratory study on retailing. *Journal of Retailing and Consumer Services*, *30*, 96-104.
- Clowes, A., Mitchell, P., & Hanson, C. (2018). The business case for reducing food loss and waste: hotels. Champions 12.3. Available at: <u>https://champions123.org/wp-content/uploads/2018/04/Report Hotels The-Business-Case-for-Reducing-Food-Loss-and-Waste.pdf</u> [Accessed: 20 July 2018].
- Coşkun, A., Özbük,Y. R., M. (2020). What influences consumer food waste behavior in restaurants? An application of the extended theory of planned behavior. *Waste Management*, *117*, 170-178.
- De Felice, F., & Petrillo, A. (2014). Proposal of a structured methodology for the measure of intangible criteria and for decision making. *International Journal of Simulation and Process Modelling*, *9*(*3*), 157-166.
- De Moraes, C.C., de Oliveira Costa, F.H., Pereira, C.R., da Silva, A.L. and Delai, I., 2020. Retail food waste: mapping causes and reduction practices. Journal of Cleaner Production, 256, p.120124.

- Derqui, B., Fayos, T., & Fernandez, V. (2016). Towards a more sustainable food supply chain: Opening up invisible waste in food service. *Sustainability*, 8(7), 693
- Duursma, G., Vrenegoor, F., & Kobus, S. (2016). Food waste reduction at restaurant De Pleats: Small steps for mankind. *Research in Hospitality Management*, 6(1), 95-100.
- Ellison, B., Savchenko, O., Nikolaus, C.J., Duff, B. R.L. (2019). Every plate counts: Evaluation of a food waste reduction campaign in a university dining hall. Resources, *Conservation and Recycling*, 144, 276-284.
- Falasconi, L., Vittuari, M., Politano. A., & Segrè, A. (2015). Food waste in school catering: An Italian case study. Sustainability, 7(11), 14745-14760.
- Fallahpour A.R., & Moghassem A.R. (2012). Evaluating applicability of VIKOR method of multicriteria decision making for parameters selection problem in rotor spinning. *Fiber Polym*, *13* (6), 802-808.
- FAO (2013). The food wastage footprint: Impact on natural resources: Summary report. Food and Agriculture Organization of the United Nations (FAO). Available at: <u>http://www.fao.org/docrep/018/i3347e/i3347e.pdf</u> [Accessed: 28 August 2018].
- FAO and International Food Waste Coalition, (2018). Do good: Save food! Education material package on food waste reduction in primary and secondary schools. Food and agriculture organization of the United Nations (FAO). Available at: <a href="http://www.fao.org/3/ca0995en/CA0995EN.pdf">http://www.fao.org/3/ca0995en/CA0995EN.pdf</a> [Accessed: 12 November 2019].
- Filimonau, V., de Coteau, D. A. (2019). Food waste management in hospitality operations: A critical review. *Tourism Management*, 71, 234-245.
- Filimonau, V., Fidan, H., Alexieva, O., Dragoev, S., Marinova, D., D. (2019a). Restaurant food waste and the determinants of its effective management in Bulgaria: An exploratory case study of restaurants in Plovdiv. *Tourism Management Perspectives*, 32, 100577.
- Filimonau, V., Krivcova, M., Pettit, F. (2019b). An exploratory study of managerial approaches to food waste mitigation in coffee shops. *International Journal of Hospitality Management*, *76*, 48-57.

- Filimonau, V., Matute, J., Kubal-Czerwińska, M., Krzesiwo, K., Mika, M. (2020a). The determinants of consumer engagement in restaurant food waste mitigation in Poland: An exploratory study. *Journal of Cleaner Production*, 247, 119105.
- Filimonau, V., Todorova, E., Mzembe, A., Sauer, L., Yankholmes, A. (2020b). A comparative study of food waste management in full service restaurants of the United Kingdom and the Netherlands. *Journal of Cleaner Production*, 258, 120775.
- Gandhi, S., Mangla, S.K., Kumar, P., & Kumar, D. (2016). A combined approach using AHP and DEMATEL for evaluating success factors in implementation of green supply chain management in Indian manufacturing industries. *International Journal of Logistics Research and Applications*, *19*(*6*), 537-561.
- Gao, S., Bao, J., Li, R., Liu, X., & Wu, C. (2021). Drivers and reduction solutions of food waste in the Chinese food service business. *Sustainable Production and Consumption*, *26*, 78-88.
- Govindan, K., Rajendran, S., Sarkis, J., & Murugesan, P. (2015). Multi criteria decision making approaches for green supplier evaluation and selection: A literature review. *Journal of Cleaner Production*, *98*, 66–83.
- Gursoy D. & G. Chi, C.G. (2020). Effects of COVID-19 pandemic on hospitality industry: review of the current situations and a research agenda. *Journal of Hospitality Marketing & Management*, 29(5), 527-529.
- Gustavsson, J., Cederberg, C., Sonesson, U., van Otterdijk, R., & Meybeck, A. (2011). Global food losses and food waste: Extent, causes and prevention. Food and agriculture organization of the United Nations. Available at: http://www.fao.org/docrep/014/mb060e/mb060e00.pdf [Accessed: 20 July 2018].
- Halloran, A.M.S., Clement, J., Kornum, N., Bucatariu, C., & Magid, J. (2014). Addressing food waste reduction in Denmark. *Food Policy*, 49 (1), 294-301.
- Harvey, J., Smith, A., Goulding, J., & Branco Illodo, I. (2019). Food sharing, redistribution, and waste reduction via mobile applications: A social network analysis. *Industrial Marketing Management*. In Press, corrected proof. Available online from 2 April 2019.

Hung, M.L., Yang, W.-F., Ma, H.-W., & Yang Y.-M. (2006). A novel multiobjective programming approach dealing with qualitative and quantitative objectives for environmental management. *Ecological Economics*, 56(4), 584-593.

Lewis, H., Downes, J. Verghese, K., & Young, G. (2017). *Food waste opportunities within the food wholesale and retail sectors*. Prepared for the NSW Environment Protection Authority by the Institute for Sustainable Futuresat the University of Technology Sydney.

- Lipinski, B., Hanson, G., Lomax, J., Kitinoja, L., Waite, R., & Searchinger, T. (2013). Reducing food loss and waste. Working Paper, Installment 2 of Creating a Sustainable Food Future. Washington, DC: World Resources Institute. Available at: http://citeseerx.ist.psu.edu/viewdoc/download?doi=10.1.1.360.951&rep=rep1&type=pdf [Accessed: 7 May 2018].
- Liu, K. M., Lin, S.H., Hsieh, J.C., & Tzeng G.H. (2018). Improving the food waste composting facilities site selection for sustainable development using a hybrid modified MADM model. *Waste Management*, 75, 44-59.
- Lorenz, B. A., Hartmann, M., Hirsch, S., Kanz, O., & Langen, N. (2017). Determinants of plate leftovers in one german catering company. *Sustainability*, *9*(5), 807.
- Lukic, R., Kljenak, D.V., & Jovancevic, D. (2014). Retail food waste management. *Management Research and Practice*, 6(4), 23-39.
- Mardani, A., Jusoh, A., Nor, KMD, Khalifah, Z., Zakwan, N., & Valipour, A. (2015). Multiple criteria decision making techniques and their applications a review of the literature from 2000 to 2014. *Economic Research-Ekonomska Istraživanja*, *28(1)*, 516-571.
- McAdams, B., von Massow, M., Gallant, M. & Hayhoe M.-A. (2019) A cross industry evaluation of food waste in restaurants. *Journal of Foodservice Business Research*, 22(5), 449-466.
- Mena, C., Terry, L. A., Williams, A., & Ellram, L. (2014). Causes of waste across multi tier supply networks: Cases in the UK food sector. *International Journal of Production Economics*, 152, 144-158.

- Mena, C., Yurt, O., & Adenso-Diaz, B. (2011). The causes of food waste in the supplier retailer interface: Evidences from the UK and Spain. *Resources, Conservation & Recycling*, 55(6), 648-658.
- Mohammed, A. (2020). Towards 'gresilient' supply chain management: A quantitative study. *Resources, Conservation and Recycling, 155*, 104641.
- Negus, R. (2004). Pub, bar or restaurant? Journal Retail Leisure Property, 3 (4), 326–330.
- Nujoom, R., Mohammed, A., & Wang, Q. (2019). Drafting a cost-effective approach towards a sustainable manufacturing system design. *Computers & Industrial Engineering*, 133, 317-330.
- Oprea, O.B., & Gaceu, L. (2016). Application of multiple criteria decision making (MCDM) in bakery industry. study case: Wastes and by-products. Bulletin of the Transilvania University of Braşov Series II: Forestry – Wood Industry - Agricultural Food Engineering, 9 (58, Part1), 89-94.
- Ortíz, M. A., Cómbita J.P., De la Hoz A.A., De Felice, F., & Petrillo A. (2016). An integrated approach of AHP-DEMATEL methods applied for the selection of allied hospitals in outpatient service. *International Journal of Medical Engineering and Informatics*, 8(2), 87-107.
- Oruezabala, G., & Rico, J.-C., (2012). The impact of sustainable public procurement on supplier management the case of French public hospitals. *Industrial Marketing Management*, 41 (4), 573-580.
- Oxford Economics, (2015). The economic contribution of the UK hospitality industry. A report prepared by Oxford Economics for the British Hospitality Association. Oxford Economics and British Hospitality Association. Available at: <u>http://www.bha.org.uk/wordpress/wp-content/uploads/2015/09/Economic-contribution-of-the-UK-hospitality-industry.pdf</u>
- Panisello, P. J., & Quantick, P.C. (2001). Technical barriers to Hazard Analysis Critical Control Point (HACCP). Food Control, 12(3), 165-173.
- Papargyropoulou, E., Wright, N., Lozano R., Steinberger, J., Padfield, R., & Ujange Z. (2016). Conceptual framework for the study of food waste generation and prevention in the hospitality sector. *Waste Management*, 49, 326-336.

- Parashar, S., Sood, G. and Agrawal, N. (2020). Modelling the Enablers of Food Supply Chain for Reduction in Carbon Footprint. *Journal of Cleaner Production*, 275, 122932.
- Parfitt, J., Barthel, M., & Macnaughton, S. (2010). Food waste within food supply chains: quantification and potential for change to 2050. *Philosophical Transactions of the Royal Society B*, *365*, 3065-3081.
- Parizeau, K., von Massow, M. and Martin, R., 2015. Household-level dynamics of food waste production and related beliefs, attitudes, and behaviours in Guelph, Ontario. Waste management, 35, pp.207-217.
- Pirani, S. I., & Arafat, H.A. (2016). Reduction of food waste generation in the hospitality industry. *Journal of Cleaner Production*, 132, 129-145.
- Priefer, C., Jörissen, J., & Bräutigam K.-R. (2016). Food waste prevention in Europe A cause- driven approach to identify the most relevant leverage points for action. *Resources, Conservation & Recycling*, *109*, 155-165.
- Raut, R.D., Gardas, B.B., Kharat, M. and Narkhede, B. (2018). Modeling the drivers of post-harvest losses– MCDM approach. *Computers and Electronics in Agriculture*, 154, 426-433.
- Rodgers, S. (2005). Applied research and educational needs in food service management. *International Journal of Contemporary Hospitality Management*, *17*(4), 302-314.
- Rodgers, W., Andrews, F., & Herzog, R. (1992). Quality of survey measures: A structural modeling approach. *Journal of Official Statistics*, 8, 251-275.
- Saaty, T. L. (1980). The Analytic Hierarchy Process. New York: Mcgraw-Hill International.
- Sakaguchi, L., Pak, N., & Potts, M.D. (2018). Tackling the issue of food waste in restaurants: Options for measurement method, reduction and behavioural change. *Journal of Cleaner Production*, *180*, 430-436.
- San Martín, D., Orive, M., Martinez, E., Iñarra, B., Ramos, S., González, N., Salas, A., Vázquez, L., & Zufía, J. (2017). Decision making supporting tool combining AHP method with GIS for implementing food waste valorisation strategies. *Waste and Biomass Valorization*, 8, 1555–1567.
- Santos, M.J.L.A. (2017). Food waste management-perceptions, decisions, and actions: The case of Guatemala city department restaurants. Uppsala University, Department of Earth Sciences.

- Shih, H.S. (2008). Incremental analysis for MCDM with an application to group TOPSIS. *European Journal of Operational Research*, *186*(2), 720-734.
- Si, S.L., You, X.Y., Liu, H.C., & Zhang, P. (2018). DEMATEL technique: A systematic review of the state-ofthe-art literature on methodologies and applications. *Mathematical Problems in Engineering*.
- Smithers, R. (2020). UK takeaway food waste rises during coronavirus lockdown. Guardian, 13 May 2020. Available at: <u>https://www.theguardian.com/environment/2020/may/13/uk-takeaway-food-waste-rises-during-coronavirus-lockdown</u>
- Tzeng, G. H., Chiang C.-H., & Li C.-W. (2007). Evaluating intertwined effects in e-learning programs: A novel hybrid MCDM model based on factor analysis and DEMATEL. *Expert Systems With Applications*, 32(4), 1028-1044.
- WIE (2014). *Hospitality industry food (waste) for thought*. Washington, DC. Worldwatch Institute Europe (WIE). Available at: http://worldwatcheurope.org/node/260 [Accessed: 20 July 2018].
- Wind, Y. (2006). Blurring the lines: Is there a need to rethink industrial marketing? *Journal of Business & Industrial Marketing*, 21(7), 474-481.
- WRAP (2017). Estimates of food surplus and waste arisings in the UK. WRAP. Available at: http://www.wrap.org.uk/sites/files/wrap/Estimates %20in the UK Jan17.pdf [Accessed: 27 August 2018].
- Wu, H. H., & Tsai, Y. N. (2012). An integrated approach of AHP and DEMATEL methods in evaluating the criteria of auto spare parts industry. *International Journal of Systems Science*, *43*(11), 2114-2124.
- Yazdani, M., Chatterjee, P., Zavadskas, E. K., & Zolfani S.H. (2017). Integrated QFD-MCDM framework for green supplier selection. *Journal of Cleaner Production*, 142 (4), 3728-3740.